Author's response to reviews

Title: A Randomized Pilot Study of an Intensive Postoperative Exercise Program compared with Usual Care following Primary Total Hip Arthroplasty in Subjects less than 65 Years of Age: Feasibility, Selection of Outcome Measures and Timing of Assessment

Authors:

Lauren A Beaupre PT, PhD (lbeaupre@ualberta.ca)
Edward CO Masson MD, FRCS (C) (emasson@edmontonortho.com)
Barbara J Luckhurst BSc PT (barbara.luckhurst@albertahealthservices.ca)
Orfan Arafah MD, FRCS (C) (arafah@ualberta.ca)
Gregory J O'Connor MD, FRCS (C) (goconnor@edmontonortho.com)

Version: 2 Date: 24 January 2014

Author's response to reviews: see over
24-Jan-14

To Whom It May Concern:

Please find a revised submission of an original research study to be considered for publication in the BMC Musculoskeletal Disorders entitled:

**A Randomized Pilot Study of an Intensive Postoperative Exercise Program compared with Usual Care following Primary Total Hip Arthroplasty in Subjects less than 65 Years of Age: Feasibility, Selection of Outcome Measures and Timing of Assessment**

The headings of Background and Conclusions were added.

The health research ethics board was named and ethics protocol number added.

A list of abbreviations was added.

Randomized clinical trial and health related quality of life were spelled out.

Yours truly,

Lauren Beaupre, PT, PhD

Associate Professor
Department of Physical Therapy
University of Alberta